Author’s response to reviews

Title: Whole body vibration compared to conventional physiotherapy in patients with gonarthrosis: A protocol for a randomized, controlled study

Authors:

Gregor Stein (gregor.stein@uk-koeln.de)
Christina Otto (christina.otto@uk-koeln.de)
Peter Knoell (peter.knoell@uk-koeln.de)
Peer Eysel (peer.eysel@uk-koeln.de)
Jan Siewe (jan.siewe@uk-koeln.de)
Thomas Kaulhausen (thomas.kaulhausen@uk-koeln.de)
Christoph Faymonville (christoph.faymonville@uk-koeln.de)
Kourosh Zarghooni (kourosh.zarghooni@uk-koeln.de)

Version: 2 Date: 11 May 2010

Author’s response to reviews:

Dear ladies and gentlemen,

our study has now been registered with the German clinical trial register (DRKS). Furthermore, sample size calculations and informations about the funding of our study have been added. In the abstract, the information concerning treatment groups has been expanded.